Author's response to reviews

Title: Pathological (late) fractures of the mandibular angle after lower third molar removal. A cases series.

Authors:

- tommaso cutilli (tommaso.cutilli@cc.univaq.it)
- theodora bourelaki (d.bourelaki@hotmail.com)
- secondo scarsella (scarsellas@tin.it)
- desiderio di fabio (ddifabio@tin.it)
- emanuele pontecorvi (emanuelepontecorvi@virgilio.it)
- pasqualino cargini (pcargini@gmail.com)
- luis junquera (Junquera@uniovi.es)

Version: 6 Date: 7 March 2013

Author's response to reviews: see over
L’Aquila, March 10, 2013

Editorial Team
Journal of Medical Case Reports

Dear Editorial Team,

Thank you for the suggestions. We made the changes required as follows:

1) We divided the cases in the abstract as in the case presentation (the terms added in the text appear red coloured, bold type and underlines).

2) We removed the face of the patient from figures three and five.

We hope that these changes improve our manuscript and make it suitable for publication in Your Journal.

Best Regards.

Prof. Tommaso Cutilli
CUTILLI Tommaso, BOURELAKI Theodora, CARGINI Pasqualino

PATHOLOGICAL (LATE) FRACTURES OF THE MANDIBULAR ANGLE AFTER LOWER THIRD MOLAR REMOVAL. Three news cases and review.

University of L’Aquila - Italy
Health Sciences Department
Maxillofacial Surgery
( T. Cutilli )

Corresponding Author:

Cutilli Tommaso
via della Comunità Europea, 13
67100 L’Aquila - Italy

e-mail: tommaso.cutilli@cc.univaq.it
fax: +39.0862.368547
ABSTRACT

Background. Pathological (late) fracture of the mandibular angle after third molar (M3) surgery is very rare (0.005% of the M3 removed). The cases reported in literature are 111. Method. AA describe three new cases in male patients 27, 32 and 36 years old which underwent to 3.8 surgical removal, respectively, 20, 22 and 25 days before. Results. Panoramic radiographs showed compound fractures without dislocation of fragments at sites of M3 extraction with oblique mesio-distal rhymes (unfavorable fractures). Removed teeth had following characteristics: 1) vertical variety, class II - C; 2) mesioangular variety, class I - B; 3) mesioangular variety, class II-C. Fractures were treated with open surgical reduction, internal fixation (IF) by miniplates and IMF (intermaxillary fixation).

Conclusions. The literature indicates that the risk of late fracture of the mandibular angle after surgery of M3 is double for total inclusions (Class II-III, type C) due to necessity of ostectomies more generous than in the partial inclusions. Important factor is also the anatomy of the teeth and the features of the teeth roots. The predominance of these fractures occurs in subjects with more than 25 years. The highest incidence (67.8% of cases) is found in the second and third week post-surgery. AA emphasize that before the M3 surgery it is extremely important to always provide adequate instructions to the patient in order to avoid early masticatory loads and prevent this event so that, although rare, can still occur.

Key Words: Late Mandibular Angle Fractures - Lower Third Molar Removal - Early masticatory loading